Author’s response to reviews

Title: Health status evaluation in 640 head and neck cancer survivors using EORTC QLQ-C30 and QLQ-H&N35 questionnaires

Authors:

Tsair-Fwu Lee (alfrequent@gmail.com)
Fu-Min Fang (tflee@cc.kuas.edu.tw)
Chih-Yen Chien (pichao99@gmail.com)
Hui-Ching Chuang (a092414103@cc.kuas.edu.tw)
Wen-Ling Tsai (1097405104@cc.kuas.edu.tw)
Chiu-Ching Tuan (cctuan@ntut.edu.tw)

Version: 4 Date: 13 August 2010

Author’s response to reviews: see over
Dear Editor-in-chief,

By on line submission to the international journal of *BMC HEALTH SERVICES RESEARCH*, we are hereby sending our revised manuscript entitled, "Health status evaluation in 640 head and neck cancer survivors using EORTC QLQ-C30 and QLQ-H&N35 questionnaires", for review. The report contains one manuscript, and six tables. This work has not been published in other journals or submitted to other periodicals for review. Hope you will find it acceptable for publication.

Sincerely yours,
Fu-Min Fang, M.D.Ph.D.